Author’s response to reviews

Title: Growth of a cohort of very low birth weight infants in Johannesburg, South Africa

Authors:

Cheryl A Mackay (cheryl.mackay@wits.ac.za)
Daynia E Ballot (daynia.ballot@wits.ac.za)
Peter A Cooper (peter.cooper@wits.ac.za)

Version: 5 Date: 26 April 2011

Author’s response to reviews:

To whom it may concern,

Formatting revisions have been made to the manuscript titled "Growth of a cohort of very low birth weight infants in Johannesburg, South Africa". The revised manuscript is resubmitted.

Kind regards,
Cheryl Mackay